Author’s response to reviews

Title: Lung function, asthma symptoms, and quality of life for children in public housing in Boston: a case-series analysis

Authors:

Jonathan I Levy (jilevy@hsph.harvard.edu)
L K Welker-Hood (kwelker@utmb.edu)
Jane E Clougherty (jcloughe@hsph.harvard.edu)
Robin E Dodson (rdodson@hsph.harvard.edu)
Suzanne Steinbach (suzanne.steinbach@bmc.org)
H P Hynes (hph@bu.edu)

Version: 5 Date: 3 December 2004

Author’s response to reviews:

December 3, 2004

To Whom It May Concern:

We have submitted a revised manuscript for the article titled "Lung function, asthma symptoms, and quality of life for children in public housing in Boston: a case-series analysis". We have approved all revisions made by the editors (to both the text and trimmed figures), and we have shortened the title of Table 6 to be 15 words. Please let us know if any additional formatting changes are required for this manuscript. We thank you for the opportunity to publish in Environmental Health: A Global Access Science Source.

Jonathan Levy
Assistant Professor of Environmental Health and Risk Assessment